Author's response to reviews

Title: Clostridium difficile 027 infection in Italy: where are we now and where are we going?

Authors:

Stefano Di Bella (stefano932@gmail.com)
Maria G Paglia (pagliamicromol@inmi.it)
Emma Johnson (emmas@hotmail.com)
Nicola Petrosillo (petrosillo@inmi.it)

Version: 2 Date: 21 June 2012

Author's response to reviews: see over
To *BMC Infectious Diseases*

The Editor

RE: manuscript submission.

June 21, 2012

Dear Editor:

Attached please find our manuscript “*Clostridium difficile 027 infection in Italy: where are we now and where are we going?*” to be submitted for publication. Our manuscript is an original contribution not previously published, and not under consideration for publication elsewhere. All the authors have seen and approved the submitted version of the paper. Our study confirms the emergence of CD 027 in our country and underlines the association of this strain with the increased severity of CDAD and the higher rates of recurrence episodes.

Kind regards

**The corresponding author**

Nicola Petrosillo, M.D.

2nd Infectious Diseases Division,
National Institute for Infectious Diseases “L. Spallanzani”,
Via Portuense, 292-00149 Rome, Italy.
e-mail: nicola.petrosillo@inmi.it
tel +390655170432  fax +390655170486